Author’s response to reviews

Title: Technology-dependency among patients discharged from a children's hospital: a retrospective cohort study

Authors:

Chris Feudtner (feudtner@email.chop.edu)
Nanci Larter Villareale (nanci.villareale@seattlechildrens.org)
Barbara Morray (barb.morray@seattlechildrens.org)
Virginia Sharp (ginny.sharp@seattlechildrens.org)
Ross M Hays (ross.hays@seattlechildrens.org)
John M Neff (john.neff@seattlechildrens.org)

Version: 4 Date: 6 May 2005

Author’s response to reviews:

Phone: 267/426-5032
Fax: 215/590-0426
email: feudtner@email.chop.edu

6 May 2005
The Editors
BMC Pediatrics

Dear Editors:

On behalf of my coauthors, I am yet again pleased to resubmit our manuscript, "Technology-dependency among patients discharged from a children's hospital: a retrospective cohort study," to BMC Pediatrics for consideration of publication.

For this revision, we amended the text (4th paragraph of the results section) to include a one-sentence description regarding the types of medications received by the children who were deemed to be dependent upon medications. We also performed all the of the formatting changes as requested (ie, space before citation reference number, elimination of the et al formatting in the reference list along with review of reference accuracy, elimination of the word "legend" from table and figure legends, and tighter cropping of the figures. We also proof read the entire manuscript.

Please don't hesitate to contact me with any questions or concerns. Thank you for your time and effort.

Sincerely,

Chris Feudtner, MD PhD MPH
Director of Research and Attending Physician
The Pediatric Advanced Care Team (PACT)
The Integrated Care Service (ICS)
The Children's Hospital of Philadelphia